

Sensitivity of Tropical Extreme Precipitation to Surface Warming in Aquaplanet Experiments Using a Global Nonhydrostatic Model

Alejandro Uribe, Jessica Vial, Thorsten Mauritsen

▶ To cite this version:

Alejandro Uribe, Jessica Vial, Thorsten Mauritsen. Sensitivity of Tropical Extreme Precipitation to Surface Warming in Aquaplanet Experiments Using a Global Nonhydrostatic Model. Geophysical Research Letters, 2021, 48, 10.1029/2020GL091371. insu-03726944

HAL Id: insu-03726944 https://insu.hal.science/insu-03726944

Submitted on 28 Jul2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL091371

Key Points:

- The sensitivity of tropical extreme precipitation to warming is larger than the Clausius-Clapeyron rate
- Results are more sensitive to horizontal resolution, when convection is parametrized than explicitly resolved
- Super Clausius-Clapeyron rates are mainly due to dynamical changes, but appear unrelated to changes in indicators of convective organization

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

A. Uribe, alejandro.uribe-cortes@misu.su.se

Citation:

Uribe, A., Vial, J., & Mauritsen, T. (2021). Sensitivity of tropical extreme precipitation to surface warming in aquaplanet experiments using a global nonhydrostatic model. *Geophysical Research Letters*, 48, e2020GL091371. https://doi.org/10.1029/2020GL091371

Received 11 DEC 2020 Accepted 26 MAR 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Sensitivity of Tropical Extreme Precipitation to Surface Warming in Aquaplanet Experiments Using a Global Nonhydrostatic Model

Alejandro Uribe^{1,2}, Jessica Vial^{2,3}, and Thorsten Mauritsen^{1,2}

¹University of Stockholm, Department of Meteorology, Stockholm, Sweden, ²Max Planck Institute for Meteorology, Hamburg, Germany, ³LMD/IPSL, CNRS, Sorbonne University, Paris, France

Abstract Increases of atmospheric water vapor holding capacity with temperature ($7\% \text{ K}^{-1}$ – $8\% \text{ K}^{-1}$, CC-rate) can lead to increasing extreme precipitation (EP). Observations show that tropical EP has increased during the last five decades with a rate higher than in the extratropics. Global climate models (GCM's) diverge in the magnitude of increase in the tropics, and cloud-resolving models (CRM's) indicate correlations between changes in tropical EP and organization of deep convection. We conducted global-scale aquaplanet experiments at a wide range of resolutions with explicit and parameterized convection to bridge the gap between GCM's and CRM's. We found increases of tropical EP beyond the CC rate, with similar magnitudes when using explicit convection and parametrized convection at the resolution it is tuned for. Those super-CC rates are produced due to strengthening updrafts where extreme precipitation occurs, and they do not exhibit relations with changes in convective organization.

Plain Language Summary Theory and observations indicate tropical extreme precipitation might increase with global warming. Projections from climate models agree on increases in the extratropics, but not in the tropics. More idealized simulations indicate links between increases of tropical extreme precipitation and changes in the spatial organization of the meteorological systems producing those extremes. Using a novel model approach, we found that tropical extreme precipitation increases with warming more than expected due to increases in the dynamics of the extreme precipitation systems, whereas changes in the spatial organization have a small role.

1. Introduction

The Clausius-Clapeyron relation provides a theoretical starting point for understanding the response of extreme precipitation to a warming climate. At lower tropospheric temperatures, this relation predicts a saturation specific humidity change of approximately 7% K^{-1} –8% K^{-1} (CC rate) (Trenberth et al., 2003). With such an increase of saturation water vapor in the atmosphere, it is likely that the amount of precipitation from events where most of the water vapor precipitates out will increase with warming, and thus this value represents a basic scaling for the sensitivity of extreme precipitation to warming (Berg et al., 2013). This behavior of extreme precipitation events stands in contrast with global mean changes in precipitation with warming, which is instead linked to enhanced longwave cooling of the troposphere. This cooling is mostly balanced by enhanced convective heating through release of latent heat, leading to increasing mean global precipitation at a rate lower than, and not physically related to, the CC rate (Boer, 1993; Mitchell et al., 1987; Newell et al., 1975).

Observations provide evidence partly in favor of this scaling of extreme precipitation. Since the 1950s, there have been statistically significant increases in the number of extreme precipitation events in more regions than there have been statistically significant decreases (Hartmann et al., 2013). However, observations indicate that sensitivities depend also on the type of precipitation with higher values than CC rate (super-CC) for convective precipitation (Berg et al., 2013). This discrepancy arises naturally between midlatitudes, where extreme precipitation is usually associated with frontal activity and midlatitude storms (Kodama et al., 2019), and the tropics where convection is the main driver. This was shown by O'Gorman (2015), who found that daily extremes in the tropics are more sensitive to climate warming than those in the extratropics and suggested one possible cause is from dynamical origin changes in vertical motion.

General Circulation Model (GCM) simulations also produce a general increase of extremes, the strength of which depends on latitude (O'Gorman, 2012; O'Gorman & Schneider, 2009b). In models from the Coupled Model Intercomparison Project (CMIP) Phase 3, for example, extratropical sensitivities consistently predict that precipitation extremes increase more slowly with surface air temperature than atmospheric water vapor content; however, tropical changes are not consistent among models, with sensitivities ranging from 1.3% K^{-1} to 30% K^{-1} (O'Gorman & Schneider, 2009a). These studies suggest that the discrepancy in the tropics may arise from inaccurate simulation of upward velocity during convection. Bhattacharya et al. (2017) suggested that to improve modeled tropical precipitation extremes, it is essential to better represent the upward velocity associated with those extremes. Increasing horizontal resolution may be a way to improve the simulation of convection in GCM's where precipitation is not resolved by the coarse grid and has to be parametrized. However, those convective parametrization schemes are sensitive to horizontal model resolution and time-step length (Benedict et al., 2017; Li, Collins, Wehner, Williamson, & Olson, 2011; Li, Collins, Wehner, Williamson, Olson, & Algieri, 2011; Lu et al., 2014; Williamson, 2013; Yang et al., 2014) and thus the sensitivity of extreme precipitation to warming varies not just among individual models, but also across horizontal resolutions with a single model.

Given the long-standing structural uncertainties among CGM's, Cloud Resolving Models in idealized setups of Radiative-Convective Equilibrium (RCE) (Manabe & Strickler, 1964; Nakajima & Matsuno, 1988; Tompkins & Craig, 1998) have been used to study tropical convection and sensitivities of extreme precipitation to warming. On such setups and under certain conditions, RCE can become unstable (Nilsson & Emanuel, 1999) and lead to spontaneous spatial organization of convection. In RCE simulations, it has been shown that extreme precipitation increases close to the CC rate if self-aggregation is absent (Muller et al., 2011; Romps, 2011) or if the degree of organization does not change (Bao et al., 2017); whereas super-CC behavior has been found when self-aggregation increases with warming (Bao & Sherwood, 2019; Bao et al., 2017; Pendergrass et al., 2016; Singleton & Toumi, 2013).

Here, we use a less idealized set of aquaplanet simulations to study the uncertainties of changes in tropical extreme precipitation by using a nonhydrostatic atmospheric GCM in rotating configuration with a meridional gradient of temperature (Medeiros et al., 2015; Neale & Hoskins, 2000). Model description and methods are presented in Section 2. In Section 3, we compare the sensitivity of tropical extreme precipitation to warming between simulations with parametrized and explicit convection and its resolution dependency, study contributors to those sensitivities, and look for relationships between the change of convective organization and the change of precipitation extremes. Finally, we present the conclusions in Section 4.

2. Model Setup and Methods

Simulations were performed using the ICOsahedral Nonhydrostatic Atmospheric general circulation model (ICON-A). ICON-A is built using the Max Planck Institute physical parametrization package, which originates from the ECHAM6.3 general circulation model (Mauritsen et al., 2019) and with adaptations to account for the change in the dynamical core and a new turbulence parametrization. A full description is given in Giorgetta et al. (2018).

The experiments were conducted using the aquaplanet configuration, which uses the Qobs zonally symmetric sea surface temperatures (SST) as surface boundary conditions (Neale & Hoskins, 2000). Owing to its simplicity (e.g., no topography, land-sea contrasts, surface heterogeneities), this configuration helps to understand the physical atmospheric processes driving the changes of extremes in response to global warming (Li, Collins, Wehner, Williamson, & Olson, 2011). Moreover, because diurnal insolation and the radiatively active species are held at equinoctial and hemispherically symmetric geometry, the model statistics are zonally and hemispherically symmetric, which helps to identify significant signals using relatively short integrations.

A range of simulations at different horizontal resolutions were performed with parametrization of convection on and off (explicit convection) for the control and uniformly increased SST of 4K (Table 1). We used explicit convection at resolutions lower than 10 km since even at coarse resolutions without parametrization of convection, models are able to produce large scale features related to convection (Retsch et al., 2019; Webb et al., 2015). The simulations were initialized analytically and used time-invariant boundary conditions for

Table 1

Experiment Resolutions, Time Step Lengths and Cases With Parameterized and Explicit Convection. The Resolution Is the Approximate Side Length of Squares With the Same Area as the Average Triangle in the ICON Grid

Grid name	Resolution (km)	Time step (min)	Parametrized	Explicit
R2B4	160	7.5	yes	yes
R2B5	80	3.75	yes	yes
R2B6	40	0.83	yes	yes
R2B8	10	0.25	no	yes

SST, spectral solar irradiation, well mixed greenhouse gases CO_2 , CH_4 , N_2O , CFCs, and O_3 concentration. All experiments were run for four years and we treated the first year as spinup, except for R2B8 where the simulation length was limited to six months with one month of spinup.

Daily zonal mean precipitation for explicit and parametrized experiments are shown in supporting information Figure S1. As is to be expected, the global mean precipitation increases with warming, whereas the meridional distribution of precipitation follows a double Intertropical Convergence Zone (ITCZ) structure for explicit convection simulations. When resolution is increased, precipitation tends to be more meridionally distributed with a displacement of the ITCZ away from the equator and an increase of midlatitude precipitation at the expense of tropical precipitation, particularly for R2B6 and R2B8. In general, however, the

meridional distribution of precipitation in the parametrized convection experiments is somewhat erratic across resolutions, and the ITCZ behaviors in our experiments differ from those of Retsch et al. (2019), who used an earlier version of ICON-A and found a single ITCZ structure for explicit convection simulations at resolutions R2B4, R2B5, and R2B6, while a double ITCZ prevailed for parametrized convection. Since extremes in the tropics are more likely to occur within the ITCZ, simulated shifts in the large-scale tropical circulation might obscure our results and so in the following we shall discuss precipitation extremes in the entire tropics from 30°S and 30°N.

To study changes in tropical Extreme Precipitation (EP) with warming, we define EP as the cases of grid points between 30°S and 30°N over the entire period exceeding the *i*th percentile of daily precipitation:

$$EP(i) = \frac{1}{n} \sum_{0}^{n} Pr_n \ge Pr_i, \qquad (1)$$

where *i* varied from 99.9 to 99.99, *Pr* is daily precipitation and *Pr_i* is the *i*th percentile of daily precipitation. The selection of this metric allows us to capture the behavior of the precipitation distribution tail instead of focusing on one particular percentile. With it, we calculate the sensitivity of tropical extreme precipitation to warming (SEP) as the fractional change in EP (δ EP(i)/EP(i)) normalized by change in temperature (δ T):

$$SEP(i) = \frac{\delta EP(i)}{\delta T \cdot EP(i)} = \frac{EP(i)_{4K} - EP(i)_{CTL}}{4K \cdot EP(i)_{CTL}},$$
(2)

where the subscripts CTL and 4K denote control and 4K experiments, respectively.

3. Results

3.1. Sensitivity of Tropical Extreme Precipitation to Warming

Sensitivities of tropical extreme precipitation to warming are displayed as solid lines in Figure 1. Extremes from both explicit and parametrized convection experiments increase with warming. For explicit convection simulations, sensitivities vary from 9% K^{-1} to 15% K^{-1} with a tendency to converge for the strongest extremes. This tendency is not clear for the experiments with parametrized convection, where at low resolution (R2B4) we observe a nearly constant sensitivity value, then at R2B5, it ranges from 14% K^{-1} to 24% K^{-1} and finally for the highest resolution (R2B6) it drops for the strongest extremes.

Fractional changes of lower tropospheric saturation specific humidity with warming in the tropics fall close to the CC rate (dots in Figure 1). In all simulations, extreme increases are higher than those changes in lower tropospheric saturation specific humidity (super-CC, compare solid lines and dots). This suggests that not just increases in the capacity of the atmosphere to hold water vapor have an impact on extremes, but other processes contribute too. However, we note that for parametrized convection the amplitude of this difference varies with resolution more than it does for explicit convection. When aggregating the simulations to the coarsest resolution (R2B4), similar tendencies are observed for explicit convection, whereas results





Figure 1. Solid lines show sensitivities of tropical extreme precipitation to warming. Fractional changes of lower tropospheric saturation specific humidity for each simulation are displayed as dots and the dashed line shows the mean across resolutions.

from parametrized convection at R2B5 and R2B6 are smaller in magnitude showing sensitivities higher and close to the CC rate, respectively (Figure S7). Since the convective parametrization implemented in ICON-A has been tuned for a grid resolution of R2B4, spurious behaviors at resolutions R2B5 and R2B6 resolutions might occur. This is mainly because the convective scheme is tuned to remove convective instability on a certain timescale. When resolution is increased from R2B4, the model dynamics may produce finer scale instabilities more rapidly than the parametrization can remove these, resulting in explicitly resolved updrafts or so-called grid-point storms (Williamson, 2013). We speculate that the erratic behavior of the higher resolution simulations is caused by the inadvertent competition between parameterized convection and partially resolved convective clouds. Increases of extreme precipitation from parametrized convection at the resolution it is tuned for (R2B4) are similar to the increases with explicit convection, particularly to that at R2B4 and one might think that the model shows a low sensitivity of extreme precipitation changes with warming to the activation of convective parametrized convection observed in Figure S1 suggest this may be coincidental.

3.2. Contributors to Increases of Tropical Extreme Precipitation

Next, we will use a scaling derived by Muller et al. (2011) that, as explained in Supporting Information S1, is valid for explicit convection simulations. The scaling separates changes in extremes in terms of changes in dynamics, through vertical motion (here pressure velocity, ω), in thermodynamics, through the vertical gradient of saturation specific humidity, and in precipitation efficiency:

$$SEP \approx \underbrace{\begin{bmatrix} \delta(\omega) \frac{\partial q_s}{\partial p} \end{bmatrix}}_{\substack{\delta T \cdot \begin{bmatrix} \omega \frac{\partial q_s}{\partial p} \end{bmatrix}}_{\text{Dynamic}}} + \underbrace{\begin{bmatrix} \omega \delta \frac{\partial q_s}{\partial p} \end{bmatrix}}_{\substack{T \cdot \begin{bmatrix} \omega \frac{\partial q_s}{\partial p} \end{bmatrix}} + \frac{\delta \epsilon}{\underbrace{\delta T \cdot \epsilon}_{Efficiency}}.$$
(3)

From the components of the scaling, and since increases of vertical gradient of saturation specific humidity with warming are expected to follow the CC rate due to the strongest vertical gradients of saturation specific





Figure 2. Sensitivity of tropical extreme precipitation to warming (black) and contributions to sensitivities from dynamics (yellow), thermodynamics (red), and precipitation efficiency (blue). Values are calculated by averaging the extremes interval (99.9–99.99). The precipitation efficiency contribution is estimated as the residual between the first two terms in the right side of Equation 3, and the actual sensitivity. The dashed gray line indicates the mean fractional change of lower tropospheric saturation specific humidity across simulations.

humidity locate in the lower troposphere, we can identify that increases of tropical extreme precipitation beyond the CC-scaling are caused by positive net contributions from dynamics and precipitation efficiency, here calculated as the residual.

Results of the scaling (Equation 3) are displayed in Figure 2. We note a similar behavior across resolutions with explicit convection, in that increases of extreme precipitation result from both positive dynamics and thermodynamics contributions. In all simulations, the dynamic contribution is positive and larger in magnitude than that from precipitation efficiency, which in all cases is negative. Thus, the combined effect of dynamics and precipitation efficiency changes contributes to the super-CC behavior of the model, whereas the thermodynamic contribution alone is close to the CC rate. Similar results are obtained for coarse grained resolutions to R2B4, see Figure S8.

3.3. Are Sensitivities Related to Changes in Convective Organization?

We showed that tropical extreme precipitation increases with warming at rates higher than the CC rate and that strengthening of convective circulations when extremes occur leads to those super-CC tendencies in our experiments with explicit convection. As mentioned in Section 1, the amplitude of the sensitivity of tropical extreme precipitation to warming might be related to changes of convective organization, whereby super-CC changes in extremes are correlated with increases of convective organization, and so we investigate if this is also the case in our simulations, including R2B4 with parametrized convection. To this end, we quantify the degree of convective organization in the tropics and its change with warming in a variety of ways: using subsiding fraction prime (SF', Noda et al., 2019) in the entire tropical band, as well as in smaller subdomains of varying sizes centered at the extreme event, and solely in the tropical band using organization index with eight-point connectivity to include coarse resolutions (Iorg_0, Becker & Wing, 2020). A detailed description of the metrics is given in Supporting Information S2.





Figure 3. Fractional changes of organization metrics for large scale (SF'_trop, Iorg__0_trop, and Iorg_trop) and for subdomains centered where extremes occur (SF'_30, SF'_10, SF'_8, SF'_6, SF'_4, and SF'_2). The vertical line separates explicit convection experiments from parametrized convection at R2B4. Note that given the resolution of R2B4, a subdomain of $2^{\circ} \times 2^{\circ}$ will contain a unique grid point and the fractional change of SF' will be zero; and subdomains from $6^{\circ} \times 6^{\circ}$, and $8^{\circ} \times 8^{\circ}$ will contain the same amount of grid points, producing equal fractional changes.

We found both increased and decreased tendencies of convective organization with warming (Figure 3). At high resolution (R2B5-8), all indices show a reduction in convective organization, whereas at coarse resolution (R2B4) convection tends to self-organize with warming in the areas where extremes occurs, while disorganize at large scale (we obtained similar results for coarse grained resolutions to R2B4, see Figure S9). Those results are opposite to Bao et al. (2017), Singleton and Toumi (2013), and Pendergrass et al. (2016) where increases of extremes correlate with self-organization. This discrepancy across resolutions suggests that changes in tropical convective organization have a negligible impact on changes of tropical extreme precipitation with warming in our simulations.

4. Conclusions

Aquaplanet simulations with the ICON-A model are performed to explore the sensitivity of tropical extreme precipitation to warming across a wide-range spatial resolutions with and without parametrization of convection. We find positive sensitivities with amplitudes larger than the increase of lower tropospheric saturation specific humidity, or CC rate, at all resolutions. Results from explicit convection simulations converge for the strongest precipitation extremes, whereas for parameterized convection simulations, the sensitivities strongly vary with horizontal resolution, although results from R2B4 are similar to those from the explicit convection simulations. We suggest this occurs since the parametrization scheme used in ICON-A was tuned for that particular resolution.

We next investigate whether dynamical changes can explain the super-CC behavior of tropical extreme precipitation in our explicit convection simulations, using a diagnostic framework. In all simulations, we

find positive contributions from dynamics, resulting in stronger updrafts where extreme precipitation occurs. Nevertheless, thermodynamical changes resulting from changes in the vertical gradient of saturation specific humidity also contributes relative to the simple CC-scaling in the higher resolution simulations, but not in the coarse resolution simulations (R2B4). Furthermore, for all simulations, reductions in precipitation efficiency with warming counteract the increases produced by dynamics and thermodynamics, but its magnitude is not enough to prevent the super-CC tendencies.

Finally, we explore whether convective organization could be involved using an array of indices that in various ways characterize the degree of organization. We find somewhat surprisingly in most cases organization decreases with warming: in the explicit convection simulations with 10–80 km resolution (R2B5-8) it decreases by about 1% K^{-1} . In the two simulations with 160 km (R2B4) convection disorganizes at large scale; but self organizes in the areas where extremes occur. We suggest more investigation is needed to better understand the processes controlling the change of organization of convection in simulations like those here presented. It is concluded that convective organization played either no or a negligible role in causing the model's super-CC behavior of tropical extreme precipitation.

Data Availability Statement

The primary data for this study, that is, the ICON-A model is described in Giorgetta et al. (2018) and its source code and components are available on https://mpimet.mpg.de/en/science/modeling-with-icon/ code-availability. The configuration scripts to run the model, the data and scripts to reproduce the figures can be found in Uribe et al. (2020).

Acknowledgments

The authors are grateful to Caroline Muller, Addisu Semie, Allison Wing, and Frida Bender who provided useful comments and suggestions. They would like to thank Clare Flynn for her helpful English revision. The study was supported by the Max Planck Society for the advancement of science, the European Research Council (highECS [grant no. 770765] and CONSTRAIN [grant no. 820829]). Computational resources were made available by Deutsches Klimarechenzentrum (DKRZ) through support from Bundesministerium für Bildung und Forschung (BMBF) and by the Swedish National Infrastructure for Computing (SNIC) at the PDC Center for High Performance Computing, KTH Royal Institute of Technology.

References

- Bao, J., & Sherwood, S. C. (2019). The role of convective self-aggregation in extreme instantaneous vs. daily precipitation. Journal of Advances in Modeling Earth Systems, 11, 19–33. https://doi.org/10.1029/2018MS001503
- Bao, J., Sherwood, S. C., Colin, M., & Dixit, V. (2017). The robust relationship between extreme precipitation and convective organization in idealized numerical modeling simulations. *Journal of Advances in Modeling Earth Systems*, 9, 2291–2303. https://doi. org/10.1002/2017MS001125
- Becker, T., & Wing, A. A. (2020). Understanding the extreme spread in climate sensitivity within the radiative-convective equilibrium model intercomparison project. Journal of Advances in Modeling Earth Systems, 12, e2020MS002165. https://doi.org/10.1029/2020MS002165
- Benedict, J. J., Medeiros, B., Clement, A. C., & Pendergrass, A. G. (2017). Sensitivities of the hydrologic cycle to model physics, grid resolution, and ocean type in the aquaplanet community atmosphere model. *Journal of Advances in Modeling Earth Systems*, 9, 1307–1324. https://doi.org/10.1002/2016MS000891
- Berg, P., Moseley, C., & Haerter, J. O. (2013). Strong increase in convective precipitation in response to higher temperatures. Nature Geoscience, 6, 181–185. https://doi.org/10.1038/ngeo1731
- Bhattacharya, R., Bordoni, S., & Teixeira, J. (2017). Tropical precipitation extremes: Response to SST-induced warming in aquaplanet simulations. *Geophysical Research Letters*, 44, 3374–3383. https://doi.org/10.1002/2017GL073121
- Boer, G. J. (1993). Climate change and the regulation of the surface moisture and energy budgets. Climate Dynamics, 8, 225–239. https:// doi.org/10.1007/BF00198617
- Giorgetta, M. A., Brokopf, R., Crueger, T., Esch, M., Fiedler, S., Helmert, J., et al. (2018). ICON-A, the atmosphere component of the ICON earth system model: I. Model description. *Journal of Advances in Modeling Earth Systems*, *10*, 1613–1637. https://doi.org/10.1029/2017MS001242
- Hartmann, D., Tank, A. K., Rusticucci, M., Alexander, L., Brönnimann, S., Charabi, Y., et al. (2013). Observations: Atmosphere and surface. In Climate change 2013: The physical science basis. Contribution of working group I to the fifth assessment report of the intergovernmental panel on climate change (p. 1535). Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- Kodama, C., Stevens, B., Mauritsen, T., Seiki, T., & Satoh, M. (2019). A new perspective for future precipitation change from intense extratropical cyclones. *Geophysical Research Letters*, 46, 12435–12444. https://doi.org/10.1029/2019GL084001
- Li, F., Collins, W. D., Wehner, M. F., Williamson, D. L., & Olson, J. G. (2011). Response of precipitation extremes to idealized global warming in an aqua-planet climate model: Towards a robust projection across different horizontal resolutions. *Tellus A: Dynamic Meteorology* and Oceanography, 63, 876–883. https://doi.org/10.1111/j.1600-0870.2011.00543.x
- Li, F., Collins, W. D., Wehner, M. F., Williamson, D. L., Olson, J. G., & Algieri, C. (2011). Impact of horizontal resolution on simulation of precipitation extremes in an aqua-planet version of Community Atmospheric Model (CAM3). *Tellus A: Dynamic Meteorology and Oceanography*, 63, 884–892. https://doi.org/10.1111/j.1600-0870.2011.00544.x
- Lu, J., Ruby Leung, L., Yang, Q., Chen, G., Collins, W. D., Li, F., et al. (2014). The robust dynamical contribution to precipitation extremes in idealized warming simulations across model resolutions. *Geophysical Research Letters*, 41, 2971–2978. https://doi. org/10.1002/2014GL059532
- Manabe, S., & Strickler, R. F. (1964). Thermal equilibrium of the atmosphere with a convective adjustment. *Journal of the Atmospheric Sciences*, 21, 361–385. https://doi.org/10.1175/1520-0469(1964)021<0361:teotaw>2.0.co;2
- Mauritsen, T., Bader, J., Becker, T., Behrens, J., Bittner, M., Brokopf, R., et al. (2019). Developments in the MPI-M earth system model version 1.2 (MPI-ESM1.2) and its response to increasing CO₂. Journal of Advances in Modeling Earth Systems, 11, 998–1038. https://doi.org/10.1029/2018MS001400

- Medeiros, B., Stevens, B., & Bony, S. (2015). Using aquaplanets to understand the robust responses of comprehensive climate models to forcing. *Climate Dynamics*, 44, 1957–1977. https://doi.org/10.1007/s00382-014-2138-0
- Mitchell, J., Wilson, C., & Cunnington, W. (1987). On Co₂ climate sensitivity and model dependence of results. *Quarterly Journal of the Royal Meteorological Society*, 113, 293–322. https://doi.org/10.1002/qj.4971134751710.1256/smsqj.47516
- Muller, C. J., O'Gorman, P. A., & Back, L. E. (2011). Intensification of precipitation extremes with warming in a cloud-resolving model. Journal of Climate, 24, 2784–2800. https://doi.org/10.1175/2011JCLI3876.1
- Nakajima, K., & Matsuno, T. (1988). Numerical experiments concerning the origin of cloud clusters in the tropical atmosphere. Journal of the Meteorological Society of Japan, 66, 309–329. https://doi.org/10.2151/10.2151/jmsj1965.66.2_309
- Neale, R., & Hoskins, B. J. (2000). A standard test for AGCMs including their physical parametrizations: I: The proposal. Atmospheric Science Letters, 1, 101–107. https://doi.org/10.1006/asle.2000.0022
- Newell, R. E., Herman, G. F., Gould-Stewart, S., & Tanaka, M. (1975). Decreased global rainfall during the past Ice Age. *Nature*, 253, 33–34. https://doi.org/10.1038/253033b0
- Nilsson, J., & Emanuel, K. A. (1999). Equilibrium atmospheres of a two-column radiative-convective model. Quarterly Journal of the Royal Meteorological Society, 125, 2239–2264. https://doi.org/10.1002/qj.49712555814
- Noda, A. T., Kodama, C., Yamada, Y., Satoh, M., Ogura, T., & Ohno, T. (2019). Responses of clouds and large-scale circulation to global warming evaluated from multidecadal simulations using a global nonhydrostatic model. *Journal of Advances in Modeling Earth Systems*, 11, 2980–2995. https://doi.org/10.1029/2019MS001658
- O'Gorman, P. A. (2012). Sensitivity of tropical precipitation extremes to climate change. Nature Geoscience, 5. https://doi.org/10.1038/ ngeo1568
- O'Gorman, P. A. (2015). Precipitation extremes under climate change. Current Climate Change Reports, 1, 49–59. https://doi.org/10.1007/ s40641-015-0009-3
- O'Gorman, P. A., & Schneider, T. (2009a). The physical basis for increases in precipitation extremes in simulations of 21st-century climate change. *Proceedings of the National Academy of Sciences*, *106*, 14773–14777. https://doi.org/10.1073/pnas.0907610106
- O'Gorman, P. A., & Schneider, T. (2009b). Scaling of precipitation extremes over a wide range of climates simulated with an idealized GCM. *Journal of Climate*, 22, 5676–5685. https://doi.org/10.1175/2009JCL12701.1
- Pendergrass, A. G., Reed, K. A., & Medeiros, B. (2016). The link between extreme precipitation and convective organization in a warming climate: Global radiative-convective equilibrium simulations. *Geophysical Research Letters*, 43, 445–511. https://doi. org/10.1002/2016GL071285
- Retsch, M. H., Mauritsen, T., & Hohenegger, C. (2019). Climate change feedbacks in aquaplanet experiments with explicit and parametrized convection for horizontal resolutions of 2,525 up to 5 km. *Journal of Advances in Modeling Earth Systems*, 11, 2070–2088. https://doi.org/10.1029/2019MS001677
- Romps, D. M. (2011). Response of tropical precipitation to global warming. Journal of the Atmospheric Sciences, 68, 123–138. https://doi. org/10.1175/2010JAS3542.1
- Singleton, A., & Toumi, R. (2013). Super-Clausius-Clapeyron scaling of rainfall in a model squall line. *Quarterly Journal of the Royal Mete*orological Society, 139, 334–339. https://doi.org/10.1002/qj.1919
- Tompkins, A. M., & Craig, G. C. (1998). Radiative-convective equilibrium in a three-dimensional cloud-ensemble model. *Quarterly Journal* of the Royal Meteorological Society, 124, 2073–2097. https://doi.org/10.1002/qj.49712455013
- Tompkins, A. M., & Semie, A. G. (2017). Organization of tropical convection in low vertical wind shears: Role of updraft entrainment. Journal of Advances in Modeling Earth Systems, 9, 1046–1068. https://doi.org/10.1002/2016MS000802
- Trenberth, K. E., Dai, A., Rasmussen, R. M., & Parsons, D. B. (2003). The changing character of precipitation. Bulletin of the American Meteorological Society, 84, 1205–1218. https://doi.org/10.1175/BAMS-84-9-1205
- Uribe, A., Vial, J., & Mauritsen, T. (2020). Replication data for experiments and figures in: Sensitivity of tropical extreme precipitation to surface warming in aquaplanet experiments using a global nonhydrostatic model. Zenodo. Retrieved from https://doi.org/10.5281/ zenodo.4674282
- Webb, M. J., Lock, A. P., Bretherton, C. S., Bony, S., Cole, J. N. S., Idelkadi, A., et al. (2015). The impact of parametrized convection on cloud feedback. *Philosophical Transactions of The Royal Society A*, 373, 20140414. https://doi.org/10.1098/rsta.2014.0414
- Williamson, D. L. (2013). The effect of time steps and time-scales on parametrization suites. *Quarterly Journal of the Royal Meteorological Society*, 139, 548–560. https://doi.org/10.1002/qj.1992
- Yang, Q., Leung, L. R., Rauscher, S. A., Ringler, T. D., & Taylor, M. A. (2014). Atmospheric moisture budget and spatial resolution dependence of precipitation extremes in aquaplanet simulations. *Journal of Climate*, 27, 3565–3581. https://doi.org/10.1175/JCLI-D-13-00468.1